Author's response to reviews

Title: Hepatic Steatosis Secondary to Capecitabine: a case report

Authors:

Sheray N Chin (sheraychin@gmail.com)  
Tae K Kim (taekyoung.kim@uhn.on.ca)  
Lillian L Siu (lillian.siu@uhn.on.ca)

Version: 2 Date: 28 January 2010

Author's response to reviews:

As requested, the following revisions have been made:

- the capitalization of the title has been corrected
- Figure 1 (made up of 4 parts 1A-1D) has been submitted as a single page
- There is a single figure legend for Figure 1
- the bullets have been removed from the Abstract
- the references have been checked and do not contain 'et al'

Additional changes:

- On the title page, email address for Sheray Chin has been changed to sheray.chin@uwimona.edu.jm